Author's response to reviews

Title: Studies on antidiarrhoeal, antispasmodic and bronchodilator activities of Operculina turpethum Linn.

Authors:

Huma Shareef (phr_huma@hotmail.com)
Ghazala H Rizwani (rizwanix2@yahoo.com)
Safur Rehman Mandukhail (saifmandukhal@yahoo.com)
Naoharu Watanabe (acnwata@ipc.shizuoka.ac.jp)
Anwarul-Hassan Gilani (anwarhgilani@yahoo.com)

Version: 4 Date: 6 November 2014

Author's response to reviews: see over
Editorial comments:

- Thank you for including an ethics statement in your manuscript. However, we would ask you to confirm that your study was submitted to, and approved by, your institutional ethics committee. Please amend your ethics statement so that this is clearly declared.

Response: There is no separate committee at the University of Karachi, to address Ethical issue only, but this study was part of PhD project of Ms Huma Shareef (1st author in this manuscript), which was approved by the Board of Advanced Study and Research (BASR), University of Karachi. The BASR is comprised of senior scientists from different fields and evaluate proposals from different angles including ethical issues. The statement in the respective section of methods has been added as desired.